Author's response to reviews

Title: Mass spectrometry protein expression profiles in colorectal cancer tissue associated with clinico-pathological features of disease

Authors:

Christopher CL Liao (chienliao@yahoo.com)
Nicholas Ward (nicholasward@talktalk.net)
Simon Marsh (Simon.Marsh@colchesterhospital.nhs.uk)
Tan Arulampalam (Tan.Arulampalam@colchesterhospital.nhs.uk)
John D Norton (jnorton@essex.ac.uk)

Version: 4 Date: 5 August 2010

Author's response to reviews:

Formatting revisions made as follows:

1) The original Table 2 (parts A and B) have now been split into two separate tables (Table 2 and Table 3). Reference made to these in the text has been amended accordingly

2) Reference 27 has now been cited in the text; references are cited in numerical order

3) Underlining of text (for refereeing purposes) has now been removed

4) Vertical lines in tables have been removed

5) Figure numbers have been removed from Figures